Author's response to reviews

Title: Developing and evaluating interventions to reduce inappropriate prescribing by general practitioners of antibiotics for upper respiratory tract infections: a randomised controlled trial to compare paper-based and web-based modelling experiments

Authors:

Shaun Treweek (streweek@mac.com)
Ian W Ricketts (ricketts@computing.dundee.ac.uk)
Jill Francis (j.francis@abdn.ac.uk)
Martin Eccles (Martin.eccles@ncl.ac.uk)
Debbie Bonetti (d.bonetti@cpse.dundee.ac.uk)
Nigel Pitts (n.b.pitts@cpse.dundee.ac.uk)
Graeme MacLennan (g.maclennan@abdn.ac.uk)
Frank Sullivan (f.m.sullivan@cpse.dundee.ac.uk)
Claire Jones (cjones@computing.dundee.ac.uk)
Karen Barnett (k.barnett@cpse.dundee.ac.uk)

Version: 3 Date: 18 January 2011

Author's response to reviews: see over
Dear Editor,

Re: Implementation Science manuscript: 1720910543500656

I have uploaded a modified version of our protocol; specifically I have made some minor changes in light of completing the CONSORT checklist:

• added ‘randomised controlled trial’ to the title (p1)
• added a paragraph on blinding (p7)
• added a sentence giving details of the software used for randomisation and our use of blocks for GP invitations (p7)

• I have also made some minor affiliation changes (p1)

I’ve also uploaded the CONSORT checklist; Figure 1 remains unchanged.

I look forward to hearing from you in due course.

Yours faithfully,

Dr Shaun Treweek
Senior Lecturer, University of Dundee
Assistant Director, Tayside Clinical Trials Unit

Contact details:
Division of Clinical & Population Science and Education
University of Dundee
Kirsty Semple Way
Dundee DD2 4BF
UK

Email: streweek@mac.com
Mob: +44 777 901 6955
Fax: +44 1382 420010